

American Journal of Drug Discovery and Development

ISSN 2150-427X



www.academicjournals.com

American Journal of Drug Discovery and Development

ISSN 2150-427x DOI: 10.3923/ajdd.2017.48.53



Research Article

Two-week Treatment with Diuretics as Add on Therapy is Enough to Control the Blood Pressure in Patients Presented with Uncontrolled-resistant Hypertension

^{1,2}Marwan Salih Mohamud Al-Nimer and ²Rozhgar Faisal Ahmed

¹Department of Dentistry, College of Ausool Al-Deen University, Baghdad, Iraq ²Department of Pharmacology, College of Pharmacy, Hawler Medical University, Erbil, Iraq

Abstract

Background and Aim: Diuretics used as a first line therapy in the management of hypertension. Spironolactone is the fourth drug that indicated in the management of resistant hypertension. This study aimed to clarify the importance of using short term therapy for 2 weeks as add on therapy in the management of uncontrolled-resistant hypertension. **Materials and Methods:** A total number of 120 patients were enrolled in open label clinical study. The patients are assigned to receive 2 week therapy of a single daily dose of amiloride (5 mg) plus hydrochlorothiazide (50 mg) (Group I), or 50 mg spironolactone (Group II) or 50 mg hydrochlorothiazide (Group III). The following measurements, including the anthropometric measurement, blood pressure, lipid profile and serum electrolytes were considered. **Results:** Diuretics produced significant reduction of the blood pressure of systolic and diastolic that amounted 23 and 14.2% (Group II), 24.3 and 14.4% (Group II) and 25.4 and 13.0% (Group III) of the baseline value. They induced alterations in serum electrolytes in Group II and III. **Conclusion:** Two- week therapeutic regimens of diuretics are enough to achieve the control of blood pressure and the logic drug for resistant hypertension is spironolactone.

Key words: Diuretics, uncontrolled resistant hypertension, short term regimen, add on therapy, serum electrolytes, lipid profile, anthropometric measurements

Received: August 17, 2016

Accepted: November 16, 2016

Published: December 15, 2016

Citation: Marwan Salih Mohamud Al-Nimer and Rozhgar Faisal Ahmed, 2017. Two-week treatment with diuretics as add on therapy is enough to control the blood pressure in patients presented with uncontrolled-resistant hypertension. Am. J. Drug Discov. Dev., 7: 48-53.

Corresponding Author: Marwan Salih Mohamud Al-Nimer, Department of Dentistry, College of Ausool Al-Deen University, Baghdad, Iraq

Copyright: © 2017 Marwan Salih Mohamud Al-Nimer and Rozhgar Faisal Ahmed. This is an open access article distributed under the terms of the creative commons attribution License, which permits unrestricted use, distribution and reproduction in any medium, provided the original author and source are credited.

Competing Interest: The authors have declared that no competing interest exists.

Data Availability: All relevant data are within the paper and its supporting information files.

INTRODUCTION

Diuretics prescribed in the management of hypertension as primary drugs or as add-on-therapy or used in combination with other antihypertensive agents like calcium entry blockers or angiotensin receptor blockers. These medications act at different sites of actions and they differed in their efficacy and inducing adverse reactions. In adult hypertensive patients, the diuretics effectively reduced the blood pressure as well as the risk of adverse cardiovascular complications¹. Thiazides (e.g., hydrochlorothiazide) reduced the blood pressure in a dose dependent manner and it reduced 11 mm Hg (systolic) and 5 mm Hg (diastolic) by using 50 mg day⁻¹ for eight weeks². Single low dose of chlorthalidone (6.25 mg) but not hydrochlorothiazide (12.5 mg) reduced the day time and night time blood pressure over 24 h monitoring which indicated that chlorthalidone is a useful remedy as monotherapy in management of hypertension³. Thiazide-type diuretics are useful first-line agents because they reduced the cardiovascular mortality and co-morbidities in hypertension (systolic and diastolic subtype)⁴. Long-term therapy with thiazides (at least more than 4 weeks) reduced the total peripheral resistance and reduced the blood pressure independent of natriuresis⁵. Loop diuretic class, as a monotherapy, reduced the blood pressure by 8 mmHg (systolic) and 4 mmHg (diastolic) after 8 weeks treatment with a fixed dose (e.g., frusemide 40-60 mg day⁻¹)⁶. This class of diuretics is preferable to treat hypertension that associated with renal impairment⁷. Potassium sparing diuretics notably spironolactone usually prescribed as the fourth drug for treatment of uncontrolled-resistant hypertension⁸.

It is estimating that 20% of hypertensive patients were resistant to at least three antihypertensive drugs and those patients have a primary aldosteronism. In one study, Verdalles *et al.*⁹ found that the resistant hypertension increased significantly with age, degree of chronic kidney disease and albuminuria and the blood pressure was controlled in about 47.5% of patients who used \geq 4 antihypertensive agents. The efficacy of amiloride (an epithelium-sodium channel blocker) in controlling the resistant form of hypertension is inferior to the spironolactone¹⁰. The rationale of this study that diuretics usually indicated as a first line therapy in the management of hypertension and their effects observed after 4 weeks but they can reduce the blood pressure within a short period with variable efficacy and minimal adverse reactions.

Therefore, this study aimed to clarify the benefit of using different modalities of diuretics as 'add on therapy' and not as first line therapy for two weeks to treat uncontrolled hypertensive patients used antihypertensive drugs other than diuretics.

MATERIALS AND METHODS

This study was reviewed by the Scientific Committee at the Department of Pharmacology and approved by the Council of College of Pharmacy at Hawler Medical University in Kurdistan region in Iraq as a part of postgraduate study. Each patient was asked to enroll in this study, according to the ethical guideline that approved by the General Directorate of Health, Erbil. The patients who are eligible for the study are adult hypertensive patients of both genders. The criteria of inclusion are hypertensive patients treated with antihypertensive drugs. Therefore, the patients included in this study were on the different antihypertensive modalities gave a picture of the variability in the number of medications at the time of entry into the study. The criteria of exclusion are end stage renal failure, emergency hypertension, preeclampsia, pregnant or lactated mother and confirmed diagnosis of secondary hypertension (polycystic kidney, Conn's syndrome and Cushing's syndrome).

The patients recruited from the Consultant Clinics of Internal Medicine at Rizgary Hospital. A total number of 120 patients who fulfilled the above criteria were enrolled in this open-labeled clinical study and they are assigned to the following treatments for two weeks according to their clinical conditions:

- Group I (n = 45): Patients treated with a combined therapy of amiloride (5 mg) plus hydrochlorothiazide (50 mg) in the pharmaceutical preparation of Moduretic tablet, single oral dose per day
- Group II (n = 44): Patients treated spironolactone (50 mg) single oral dose per day
- **Group III (n = 31):** Patients treated hydrochlorothiazide (50 mg) single oral dose per day

Each patient was clinically assessed by the consultant internal medicine before assignment to the diuretic treatment. Then each patient was subjected to the following measurements and laboratory investigations.

Blood pressure measurement: Mercury sphygmomanometer was used to measure both systolic and diastolic blood pressure manually. The patient asked to relax on chair for 5 min and then blood pressure was by the palpatory method followed by ausculatory method. The pulse and mean arterial pressure were determined by using the following formula: Pulse pressure (mmHg) = Systolic BP-diastolic BP Mean arterial BP (mmHg) = Diastolic $BP + \frac{1}{3}$ pulse pressure

Two measurements of blood pressure were recorded, the first measurement on the admission and the second measurement after two weeks of treatment.

Anthropometric measurements: The anthropometric measurements included weight (kg) and height (m). The Body Mass Index (BMI) was calculated according to the Quetlett's equation:

BMI (kg m⁻²) =
$$\frac{\text{Weight (kg)}}{\text{Height}^2 (m)}$$

Laboratory investigations: The following biochemical tests were determined in the laboratories of the hospital as a part of routine investigations of the patient's assessment. A venous blood was obtained from each patient (fasting state) by venipuncture and the serum was separated by centrifugation (3000 rpm for 5 min) and collected for determination of fasting glucose and lipid profile Total Cholesterol (TC), triglycerides (TG) and High Density Lipoprotein cholesterol (HDL-c)), uric acid, creatinine and electrolytes (sodium and potassium). Fasting serum Low Density Lipoprotein-cholesterol (LDL-c) was calculated by using the following equation:

Fasting serum LDL-c= TC- ((0.2×TG)+(HDL-c))

Serum electrolytes were determined by a direct method using i-Smart Electrolyte Analyzer (serial number 30551, Korea). The principle of this testing is electrochemistry, by which a serum sample of 60 µL was aspirated and the voltage potential of sodium and potassium was simultaneously

Table 1: Characteristics of the patients enrolled in the study

recorded and calculated accordingly to the reference values. The results expressed in mmol L^{-1} . At the time of the entry into the study, the mean levels of serum creatinine and uric acid were 0.87 and 3.91 mg d L^{-1} , respectively and no patient have abnormal levels.

Statistical analysis: The results are expressed as number, percentage and mean \pm SD. The data were analyzed using Student's t test (two tailed, paired), one way analysis of variance (ANOVA) *post hoc* Tukey (HSD) test and the difference between percentage test using XLSTAT 2015 program. The lower limit of significance is p<0.05.

RESULTS

Table 1 showed the characteristics of the patients enrolled in this study. There was insignificant differences between treated groups regarding the age, duration of hypertension, current smoking, history of taking contraceptive pills or the presence of co-morbidities e.g. diabetes mellitus or dyslipidemia. Significant high frequencies of uncontrolled hypertension (patients used < 3 antihypertensive drugs) were found in the Group I and III compared with Group II (Table 1). The patients of Group II were included uncontrolled (43.2%) and resistant (56.8%) hypertension. The cardiometabolic risk factors were observed in all patients enrolled in the study and did not show significant differences between studying groups except the mean arterial blood pressure, which attended significant low value in Group III compared with Group I and Group II (Table 2). The mean value of BMI was in the range of overweight and the mean value of fasting serum triglycerides was above the normal cutoff level of 150 mg dL⁻¹ (Table 2).

Figure 1 showed that two-week treatment with diuretics significantly reduced both systolic and diastolic blood pressures. Combined therapy of hydrochlorothiazide and

Characteristics	Group I (n = 45)	Group II ($n = 44$)	Group III (n = 31)
Gender (Male: Female ratio)	11:34	13:31	5:9
Age (year)	51.2±9.1	54.5±9.3	50.1±16.0
Current smoking	8 (17.8)	8 (18.2)	4 (28.6)
History of contraceptive pills	15 (33.3)	14 (31.8)	4 (28.6)
Duration of hypertension (year)	5.8±4.5	7.0±5.1	4.9±3.7
Number of antihypertensive			
<3 drugs	45 (100)	19 (43.2)*	14 (100)†
≥3 drugs	(0.0)	25 (56.8)	0 (0)
Co-morbidities			
Diabetes mellitus	7 (15.6)	8 (18.2)	2 (14.3)
Dyslipidemia	14 (31.1)	15 (34.1)	7 (50.0)

The results are expressed as number (percentage) and Mean±SD. Group I: Patients treated with hydrochlorothiazide and amiloride, Group II: Patients treated with spironolactone, Group III: Patients treated with hydrochlorothiazide, *p<0.001 compared with Group I, †p<0.001 compared with Group II



Fig. 1: Effect of 2 week treatment with diuretics on the blood pressure. The results are expressed as Mean±SD. Group I: Patients treated with hydrochlorothiazide and amiloride, Group II: Patients treated with spironolactone, Group III: Patients treated with hydrochlorothiazide, *p<0.001 was calculated by using paired Student's t test

Table 2: Cardio-metabolic risk factors						
Characteristics	GroupI(n=45)	Group II (n = 44)	Group III (n = 31			
Body mass index (kg m ⁻²)	28.5±3.6	28.0±3.0	27.6±2.4			
Fasting serum levels of:						
Glucose						
Total cholesterol	184.2±43.3	190.7±36.1	187.1±34.4			
Triglycerides	181.5±122.0	215.0±99.7	185.5±76.9			
High density lipoprotein cholesterol	43.8±12.5	45.5±12.2	47.9±14.5			
Low density lipoprotein cholesterol	108.8±45.4	102.0±58.2	102.0±44.2			
Blood pressure (mmHg)						
Systolic	169.4±14.6	171.4±15.5	162.9±13.3			
Diastolic	87.2±9.5	87.0±10.5	82.1±7.0			
Mean	114.6±9.5	115.1±10.8	98.8±13.8*†			

The results are expressed as Mean±SD Group I: Patients treated with hydrochlorothiazide and amiloride, Group II: Patients treated with spironolactone, Group III: Patients treated with hydrochlorothiazide, *p<0.001 compared with Group II, †p<0.001 compared with Group I

Groups of patients	Serum potassium (mmol L ⁻¹)				
	Before treatment	After treatment	Before treatment	After treatment	
Group I (n = 45)	137.7±4.4	135.2±4.6**	4.21±0.55	4.24±0.56	
Group II (n = 44)	136.4±4.4	135.9±4.2	4.20±0.35	4.53±0.73***	
Group III (n = 31)	134.8±3.7	132.9±3.3***	4.36±0.5	4.00±0.43*	

The results are expressed as Mean±SD. Group I: Patients treated with hydrochlorothiazide and amiloride, Group II: Patients treated with spironolactone, Group III: Patients treated with hydrochlorothiazide. Statistical differences were calculated by using paired Student's t test. *p<0.02, **p<0.005, ***p<0.001

amiloride reduced the systolic and the diastolic blood pressures by 23 and 14.2% of the baseline value, respectively in Group I patients. In Group II, spironolactone reduced the systolic and the diastolic blood pressures by 24.3 and 14.4% of

the baseline value, respectively. Hydrochlorothiazide reduced the systolic and the diastolic blood pressures by 25.4 and 13.0% of the baseline value, respectively. All diuretics reduced the blood pressures to the within normal levels, that is, systolic (<140 mm Hg) and the diastolic (<90 mm Hg). Two week treatment with diuretics significantly altered the serum electrolytes. Serum sodium level significantly reduced by 1.9 mmol L⁻¹ from the mean baseline level in patients of Group III (Table 3). Potassium level significantly increased by 0.33 mmol L⁻¹ from the mean baseline level in patients of Group II and significantly decreased by 0.36 mmol L⁻¹ from the mean baseline level 3).

DISCUSSION

The results of this study highlight three important observations. First, diuretics of whatever class, they reduced the blood pressure when they prescribed as add on therapy instead of using these compounds as a first line therapy. Second, the magnitude of blood pressure reduction is so high and achieved within a short period. Third, the significant alterations of the serum electrolytes do not manifest clinically. Previous studies reported that a fixed dose of the aldosterone inhibitor (amiloride) and thiazide diuretics (hydrochlorothiazide) reduced the blood pressure by 31 mmHg (systolic) and 15 mmHg diastolic after 3 weeks of treatment of resistant hypertension¹¹. Our results are in agreement with that study except Group I represented patients with uncontrolled (used <3 medications) and only one patient presented with resistant hypertension. Literature review did not disclose the benefit of such therapy as add on therapy. Corea et al.¹² reported that fixed dose of moduretic for 4 weeks did not achieve a satisfactory reduction of the blood pressure in hypertensive patients resistant to atenolol therapy. In Group II, spironolactone therapy is effective to reduce the blood pressure to the normal levels in both uncontrolled and resistant hypertension. Our results are in agreement with Oliveras et al.13 study that finds the effectiveness of spironolactone, as add on therapy for 6 months, in reducing the ambulatory 24 h systolic and diastolic blood pressure in resistant hypertension patients and such therapy was adversely affecting the glomerular filtration rate. Batterink et al.14 explored that spironolactone can reduce the blood pressure in a dose up to 50 mg day⁻¹ but the significant reduction of blood pressure in patients with a primary hypertension can achieve with $100-500 \text{ mg day}^{-1}$. In respect of duration of antihypertensive prescription, Chaturvedi et al.¹⁵ reported a range of 3-24 weeks is enough to evaluate the antihypertensive effect of a drug as a monotherapy or combined with other medications. Williams et al.¹⁶ found that spironolactone was the most effective add on drug for the treatment of resistant hypertension compared with α or β -adrenoceptor blocking

agents and it induced insignificant and occasionally increased serum potassium levels. Accordingly, our results add new information, including short term therapy a fixed dose of 50 mg is enough to reduce the blood pressure and a routine testing of serum electrolytes is essential to detect subclinical hyperkalemia. The third treatment option that used in our study is hydrochlorothiazide. There is no doubt that most hypertensive patients were preferred a combined therapy of a small dose of hydrochlorothiazide and calcium entry blockers or angiotensin receptor blockers^{17,18} or the patients used chlorthalidone¹⁹ as a first line therapy and those patients. Therefore, the use of single dose of 50 mg hydrochlorothiazide may be carried a certain limitations. The results of our study showed that hydrochlorothiazide as add on therapy is effective as other diuretics in uncontrolled hypertension while Roush and Sica19 mentioned in their review that hydrochlorothiazide was less effective than antihypertensive agents in decreasing the systemic blood pressure. The explanation for this discrepancy is due to the methodology of assessment of hypertension in which in home 24 h reduction of the blood pressure was an index of the effectiveness of antihypertensive agents in the Roush and Sica study¹⁹. The limitations of the study included small sample size of resistant hypertension, serum renin levels did not measure and 24 h changes of the blood pressure at home or in the office did not practice.

CONCLUSION

It is conclude that a therapeutic short term (only two weeks) regimen of diuretics of whatever class is enough to reduce the blood pressure to the normal levels in uncontrolled hypertension and spironolactone therapy is a logic the fourth drug that used as add on therapy to control resistant hypertension. It is necessary to measure the serum electrolyte in patients who are currently on diuretics therapy.

SIGNIFICANCE STATEMENTS

- Short-term therapy (two weeks) is enough to control high blood pressure in patients used anti-hypertensive drugs without getting the goal of control
- It is not necessary to use diuretics for long time and they can used whenever there is resistant or uncontrolled hypertension
- Diuretics of whatever mechanism of action can reduce the blood pressure without clinical adverse reactions
- It is necessary to measure the serum electrolytes when diuretics are used. The laboratory findings of serum electrolyte alteration are usually not manifested clinically

ACKNOWLEDGMENTS

The authors expressed their gratitude and thanks to the College of Pharmacy represented by the Dean (Assistant professor Dr. Alaa Alnakshabandi) and the Head of Department of Pharmacology and Toxicology (Assistant professor Dr. Ansam Alhusseiny) to give us the opportunity and of unlimited facilities of doing this research.

REFERENCES

- 1. Blowey, D.L., 2016. Diuretics in the treatment of hypertension. Pediatr. Nephrol., 31: 2223-2233.
- 2. Musini, V.M., M. Nazer, K. Bassett and J.M. Wright, 2014. Blood pressure-lowering efficacy of monotherapy with thiazide diuretics for primary hypertension. Cochrane Database Syst. Rev. 10.1002/14651858.CD003824.pub2
- Pareek, A.K., F.H. Messerli, N.B. Chandurkar, S.K. Dharmadhikari and A.V. Godbole *et al.*, 2016. Efficacy of low-dose chlorthalidone and hydrochlorothiazide as assessed by 24-h ambulatory blood pressure monitoring. J. Am. Coll. Cardiol., 67: 379-389.
- 4. Sica, D.A., B. Carter, W. Cushman and L. Hamm, 2011. Thiazide and loop diuretics. J. Clin. Hypertens., 13: 639-643.
- 5. Chan, C.Y., E.J. Peterson and T.M. Ng, 2012. Thiazide diuretics as chronic antihypertensive therapy in patients with severe renal disease-ls there a role in the absence of diuresis? Ann. Pharmacother., 46: 1554-1558.
- Musini, V.M., P. Rezapour, J.M. Wright, K. Bassett and C.D. Jauca, 2012. Blood pressure lowering efficacy of loop diuretics for primary hypertension. Cochrane Database Syst. Rev. 10.1002/14651858.CD003825.pub4
- Izzo, Jr. J.L. and S.W. Tobe, 2016. Is there a preferred diuretic class for patients with renal impairment and hypertension? J. Am. Soc. Hypertens., 10: 282-284.
- 8. Dudenbostel, T. and D.A. Calhoun, 2017. Use of aldosterone antagonists for treatment of uncontrolled resistant hypertension. Am. J. Hypertens., 30: 103-109.
- Verdalles, U., M. Goicoechea, S.G. de Vinuesa, B. Quiroga and I. Galan *et al.*, 2016. [Prevalence and characteristics of patients with resistant hypertension and chronic kidney disease]. Nefrologia, 36: 523-529, (In Spanish).

- 10. Lane, D.A. and D.G. Beevers, 2007. Amiloride 10 mg is less effective than spironolactone 25 mg in patients with hypertension resistant to a multidrug regime including an angiotensin-blocking agent. J. Hypertens., 25: 2515-2516.
- 11. Eide, I.K., P.A. Torjesen, A. Drolsum, A. Babovic and N.P. Lilledahl, 2004. Low-renin status in therapy-resistant hypertension: A clue to efficient treatment. J. Hypertens., 22: 2217-2226.
- 12. Corea, L., M. Bentivoglio and P. Verdecchia, 1983. Effects of hydrochlorothiazide combined with amiloride in atenolol-resistant hypertensive patients. Int. J. Clin. Pharmacol. Res., 3: 367-370.
- 13. Oliveras, A., P. Armario, A. Clara, L. Sans-Atxer, S. Vazquez, J. Pascual and A. de la Sierra, 2016. Spironolactone versus sympathetic renal denervation to treat true resistant hypertension: Results from the DENERVHTA study-a randomized controlled trial. J. Hypertens., 34: 1863-1871.
- Batterink, J., S.N. Stabler, A.M. Tejani and C.T. Fowkes, 2010. Spironolactone for hypertension. Cochrane Database Syst. Rev. 10.1002/14651858.CD008169.pub2
- Chaturvedi, S., D.H. Lipszyc, C. Licht, J.C. Craig and R. Parekh, 2014. Pharmacological interventions for hypertension in children. Evidence-Based Child Health: Cochrane Rev. J., 9:498-580.
- Williams, B., T.M. MacDonald, S. Morant, D.J. Webb and P. Sever *et al.*, 2015. Spironolactone versus placebo, bisoprolol and doxazosin to determine the optimal treatment for drug-resistant hypertension (PATHWAY-2): A randomised, double-blind, crossover trial. Lancet, 386: 2059-2068.
- 17. Greathouse, M.K. and M.R. Weir, 2012. The role of ARBs alone or with HCTZ in the treatment of hypertension and prevention of cardiovascular and renal complications. Postgrad. Med., 124: 40-52.
- Izzo, Jr. J.L., Y. Jia and D.H. Zappe, 2017. Influence of age and race on 24-hour ambulatory blood pressure responses to valsartan, hydrochlorothiazide and their combination: Implications for clinical practice. J. Clin. Hypertens., 19: 143-150.
- 19. Roush, G.C. and D.A. Sica, 2016. Diuretics for hypertension: A review and update. Am. J. Hypertens., 29: 1130-1137.